Author’s response to reviews

Title: Optic disc morphology in unilateral branch retinal vein occlusion using spectral domain optical coherence tomography

Authors:
Andrea Szigeti (szigandi@gmail.com)
Miklós Schneider (schneider.miklos@med.semmelweis-univ.hu)
Mónika Ecsedy (ecsedy@yahoo.co.uk)
Zoltán Zsolt Nagy (nagy.zoltan_zsolt@med.semmelweis-univ.hu)
Zsuzsanna Récsán (zsrecsan@gmail.com)

Version: 2 Date: 16 Nov 2015

Author’s response to reviews:

Dear Editor,

We would like to thank You and Reviewers for the valuable and constructive comments on our manuscript entitled “Optic disc morphology in unilateral branch retinal vein occlusion using spectral domain optical coherence tomography”

In the following, we answer to the specific comments and additional requests of Editor:

1. References

Please note that reference 16 missing from the list of references.

Thank you for your notice. We inserted the missing reference 16. (Line 428-430)


2. Supplementary tables

Please ensure that the supplementary tables S1 and S2 are explicitly referred to in the main text.
We thank the Editor for this comment. S1 and S2 are referred to in the main text. (Line 72 and 76)

3. Section titles

In order to comply with the journal format, please rename section "Materials and Methods" to only "Methods".

Thank you for this comment. We rename the section "Materials and Methods" to "Methods" according to the journal format.

We hope that our revised manuscript will meet the high demands of the Journal.

Sincerely yours,

Andrea Szigeti, MD
Semmelweis University
Department of Ophthalmology
39 Mária Str.
1085 Budapest, Hungary
Tel.: +361-4591500/4534
Fax.: +361-3179061
Email: szigandi@gmail.com